Author's response to reviews

Title: The ties that bind: an integrative framework of Physician-Hospital Alignment

Authors:

Jeroen Trybou (jeroen.trybou@ugent.be)
Paul Gemmel (paul.gemmel@ugent.be)
Lieven Annemans (lieven.annemans@ugent.be)

Version: 2 Date: 6 January 2011

Author's response to reviews:

Dear Editor, dear reviewers,

Thank you very much for providing us with your constructive comments and suggestions. We think our paper has improved a lot due to your input. The detailed feedback allowed us to improve on logic, structure and writing style. Below we provide you with our point-by-point responses.

Thank you again for all of your useful contributions.

Jeroen Trybou
Lieven Annemans
Paul Gemmel